Author's response to reviews

Title: An Interdisciplinary Knowledge Translation Intervention in Long-Term Care: Study protocol for the Vitamin D in Osteoporosis Study (ViDOS) Pilot Cluster Randomized Controlled Trial

Authors:

Courtney C. Kennedy (kennedyc@HHSC.CA)
George Ioannidis (g.ioannidis@sympatico.ca)
Lora M. Giangregorio (lora.giangregorio@uwaterloo.ca)
Jonathan D. Adachi (jd.adachi@sympatico.ca)
Lehana Thabane (thabanl@mcmaster.ca)
Suzanne N. Morin (suzanne.morin@mcgill.ca)
Richard G. Crilly (rcrilly@uwo.ca)
Sharon Marr (marrs@mcmaster.ca)
Robert G. Josse (jossr@smh.toronto.on.ca)
Lynne Lohfeld (lofeld@mcmaster.ca)
Laura E. Pickard (pickardl@HHSC.CA)
Susanne King (kingsus@HHSC.CA)
Mary-Lou van der Horst (dhm9@xplornet.com)
Glenda Campbell (gcampbell@medicalpharmacies.com)
Jackie Stroud (jstroud@medicalpharmacies.com)
Lisa Dolovich (ldolovic@mcmaster.ca)
Anna M. Sawka (amsawka@yahoo.com)
Ravi Jain (rjain@osteoporosis.ca)
Lynn Nash (nashl@mcmaster.ca)
Alexandra Papaioannou (papaioannou@HHSC.CA)

Version: 4 Date: 2 May 2012

Author's response to reviews:

The revised mailing addresses and email addresses were inserted as requested. All other copy-edited changes looked fine.

Thank-you

Courtney Kennedy